Response to Reviewers Comments

We thank the reviewers for their comments. The necessary changes have been made accordingly.

Reviewer #1

1. No “references” are given in the background section.
   Ans: References in the background section have been added.

2. Please use latest updated guidelines to define and treat and mention treatment considerations for subclinical endocrine diseases.
   Ans: Updated guidelines have been referred to and cited as follows:

   **For prediabetes** –

   **For osteopenia** –

   **For subclinical hypothyroidism** –
   a) Guideline NG145 NI. Thyroid disease: assessment and management. (NICE guideline for subclinical hypothyroidism)


For MACS –

a) European Society of Endocrinology Clinical Practice Guideline in collaboration with the European Network for the Study of Adrenal Tumors, 2016


Reviewer #2

1. Although this is a narrative unsystematic review, it is still preferable to outline briefly the method used to identify the literatures included (eg. years, keywords, search methodology used). This can provide readers a good overview of the evidence included in this manuscript. Ans: The relevant search strategy has been included in the article after the “background” section.

2. Table 3: Unit of blood sugar & HbA1C should be stated. The different definition of AHA, WHO should be referenced as in Table 2 for consistency. Ans: The definitions have been referenced and the units have been stated.

3. Some typos are noted within the text.
Ans: Corrections have been done.

Company editor-in-chief:

1. Authors are required to provide standard three-line tables Ans: Tables have been formatted accordingly

2. Authors are advised to apply a new tool, the RCA. RCA is an artificial intelligence technology-based open multidisciplinary citation analysis database. In it, upon obtaining search results from the keywords entered by the author, "Impact Index Per Article" under "Ranked by" should be
selected to find the latest highlight articles, which can then be used to further improve an article under preparation/peer-review/revision

Ans: RCA tool has been used as per suggestion and mentioned in the search strategy as well.